Author's response to reviews

Title: A school-based intervention to promote physical activity among adolescent girls: Rationale, design, and baseline data from the Girls in Sport group randomised controlled trial

Authors:

Anthony D Okely (tokely@uow.edu.au)
Wayne G Cotton (wayne.cotton@sydney.edu.au)
David R Lubans (David.Lubans@newcastle.edu.au)
Philip J Morgan (Philip.Morgan@newcastle.edu.au)
Lauren Puglisi (laurenp@nswcc.org.au)
Judy Miller (jimiller7@pobox.une.edu.au)
Jan Wright (jwright@uow.edu.au)
Marijka J Batterham (marijka@uow.edu.au)
Louisa R Peralta (louisa.peralta@sydney.edu.au)
Janine Perry (jperry@scu.edu.au)

Version: 2 Date: 18 August 2011

Author's response to reviews:

Dear Editor. We have made the requested formatting changes to the manuscript and checked it for any typographical errors. We have attached the final revised version which we believe is ready for publication.

Regards
Dr Tony Okely
Corresponding author